Author's response to reviews

Title: Risk factors for adverse perinatal outcomes in imprisoned pregnant women: a systematic review

Authors:

Marian Knight (marian.knight@npeu.ox.ac.uk)
Emma Plugge (emma.plugge@dphpc.ox.ac.uk)

Version: 3 Date: 6 October 2005

Author's response to reviews:

Formatting changes have been made as requested:
Title, abstract and author list are identical in the manuscript and submission system.
'Introduction' changed to 'Background'.
Authors contributions changed to Authors' contributions.
Tables divided into cells/fields as requested.
Figures cropped as per instructions in the online cropping guide.